Author's response to reviews

Title: Perceived connections between information and communication technology use and mental symptoms among young adults

Authors:

Sara Thomée (sara.thomee@amm.gu.se)
Lotta Dellve (lotta.dellve@amm.gu.se)
Annika Härenstam (annika.harenstam@av.gu.se)
Mats Hagberg (mats.hagberg@amm.gu.se)

Version: 3 Date: 27 November 2009

Author's response to reviews:

Dear Editors,

Please find a revised version of our manuscript Perceived connections between information and communication technology use and mental symptoms among young adults – a qualitative study submitted for publication in BMC Public Health.

We have addressed the editorial comments and requests by,

1. The manuscript has been revised through a professional copyediting service (native English) and we have accepted and carried out the suggested alterations as far as possible. We use US spelling.

2. We have added information to the quotes of interviewed participants by giving the participants unique references (Man A, Man B, Woman A etc) and have also added age of the participant.

We hope that the requests are satisfactorily addressed and that the manuscript now is acceptable for publication.

Yours sincerely

Sara Thomée

Occupational and Environmental Medicine
Sahlgrenska Academy of Public Health and Community Medicine
Gothenburg University
Box 414
405 30 Gothenburg
Sweden